Author's response to reviews

Title: Feasibility and coverage of implementing intermittent preventive treatment of malaria in pregnant women contacting private or public clinics in Tanzania: experience-based viewpoints of health managers in Mkuranga and Mufindi districts

Authors:

Godfrey M Mubyazi Dr (gmmubyazi@gmail.com)
Pascal Magnussen Dr (pma@sund.ku.dk)
Jens Byskov Dr (jby@sund.dk)
Paul Bloch Dr (pabc@steno.dk)

Version: 4 Date: 4 September 2013

Author's response to reviews:

Dear Editor,

We greatly appreciate the opportunity you have given us to revise the manuscript (MS). We also present our many thanks to the Reviewers for their constructive comments. Here are the responses on how we worked on the minor and final comments given by the last reviewer (Elisa Sicuri) which we really appreciate:

References:

We used the references cited for us to review well and were able to cite them in the relevant sections of the main text and at the end of the reference list. They have given us additional insight.

Typos and grammatical errors:

We have re-read the paper to straighten the language, by shortening the sentences where necessary after adding some texts and checked the spelling.

Thank you once again, and looking forward to seeing the paper finally being accepted for publication.

Godfrey M Mubyazi
PRINCIPAL RESEARCH SCIENTIST (&Principal Author)